Author's response to reviews

Title: Evaluation of Abelmoschus moschatus extracts for antioxidant, free radical scavenging, antimicrobial and antiproliferative activities using in vitro assays

Authors:

Mir Zahoor Gul (ziahgul@gmail.com)
L. Md. Bhakshu (bakshu@yahoo.com)
Farhan Ahmad (btfarhan@gmail.com)
Anand K Kondapi (akksl@uohyd.ernet.in)
Insaf A Qureshi (iaqsl@uohyd.ernet.in)
Irfan A Ghazi (irfan@uohyd.ernet.in)

Version: 3 Date: 8 August 2011

Author's response to reviews: see over
To
Editor-in-Chief
BMC Complementary and Alternative Medicine

Subject: Submission of final revised manuscript after Proof-Reading-Reg.

Ref: MS: 2848705425356350

Dear Sir,

Please find enclosed the revised manuscript after proof-reading of our original research for its publication in your esteemed journal “BMC Complementary and Alternative Medicine”. We have re-revised the final manuscript exactly as per suggestions and comments of given by you and also providing you a point-by-point response in the tabular form.

With Warm Regards
Yours sincerely

Irfan A. Ghazi
Assistant Professor
Department of Plant Sciences,
School of Life Sciences,
University of Hyderabad, Gachibowli,
Prof C.R. Rao Road, Hyderabad 500046. India.
Ref: MS: 2848705425356350 entitled “Evaluation of Abelmoschus moschatus extracts for antioxidant, free radical scavenging, antimicrobial and antiproliferative activities using in vitro assays.” Mir Z Gul, Lepakshi M Bhakshu, Farhan Ahmad, Anand K Kondapi Insaf A Qureshi and Irfan A Ghazi*

**Point-by-point description of the changes made**

<table>
<thead>
<tr>
<th>S.No</th>
<th>Revisions to be made</th>
<th>Remarks</th>
</tr>
</thead>
<tbody>
<tr>
<td><strong>Major Revisions</strong></td>
<td></td>
<td></td>
</tr>
<tr>
<td>1</td>
<td>Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)</td>
<td>The list of authors have been modified and written in same style and order as per suggestions given by the editor.</td>
</tr>
<tr>
<td>2</td>
<td>Structure: Please include methods section either after background or conclusion.</td>
<td>Methods section has been given after the background in the manuscript.</td>
</tr>
<tr>
<td>3</td>
<td>Figures: The first figure referenced in the manuscript text should be Figure 1. The second, figure 2, etc. Please reorder the figure citation so that they are cited in consecutive order. All figures must be cited; figures not cited will be excluded from the final manuscript.</td>
<td>Necessary changes have been made. All the figures have been cited in the manuscript.</td>
</tr>
<tr>
<td><strong>Minor Revisions</strong></td>
<td></td>
<td></td>
</tr>
<tr>
<td>4</td>
<td>References: the reference list should contain up to 30 authors names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.</td>
<td>The term ‘et al’ has been removed from the references and full list of authors have been given.</td>
</tr>
<tr>
<td>5</td>
<td>Tables: Please remove vertical lines.</td>
<td>Vertical lines have been removed as per the format of the journal.</td>
</tr>
<tr>
<td></td>
<td>Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.</td>
<td>We have once again checked the manuscript for any typographical errors. The necessary changes have been made wherever it was needed.</td>
</tr>
</tbody>
</table>